Author's response to reviews

Title: Determinants of unplanned pregnancies in rural Ghana

Authors:

Sebastian Eliason (sakeliason@yahoo.co.uk)
Frank Baiden (baidenf@yahoo.co.uk)
Barbara Yankey (hellobabs8@yahoo.com)
Kofi Awusabo-Asare (awusabo_asare@yahoo.co.uk)

Version: 3  Date: 5 January 2014

Author's response to reviews:

5TH January 2014

The Editor-in-chief
BMC Public Health Journal
UK
SUBMISSION OF RESEARCH ARTICLE TITLED “DETERMINANTS OF UNPLANNED PREGNANCIES IN RURAL GHANA”

Unplanned pregnancies in low income countries including Ghana have become a major reproductive health problem. This is because most of them end up in unsafe abortions, which are major causes of morbidity and mortality amongst women within the reproductive age group. The search for determinants will go a long way to help prevent these unplanned pregnancies. Only one paper has detailed analysis of determinants of unplanned pregnancies in Ghana. Our article elucidates very interesting determinants which will support preventive efforts by the Ghana Health Service.

We the under listed authors hereby submit for review and publication the attached original article in your esteemed journal.

This article was submitted to BMC Globalization and Health on 12th December 2013, but the Editorial team considered it unsuitable for their journal and suggested a transfer to BMC-Public Health with the reasons that the nature of the research may fit this journal better and publication may reach the right audience. No editorial review comments have been made available to me yet. We would be happy to revise the manuscript if any comments are sent to me.

This research has not been submitted to any other journal for publication and we the authors declare no conflict of interest.

Yours faithfully

Dr Sebastian Eliason
Corresponding Author/Principal Investigator

Authors:

1. Dr Sebastian Eliason, MD, MPH, FRSPH, PhDc, Department of Community Medicine, School of Medical Sciences, University of Cape Coast, Cape Coast, Ghana. Email: sakeliason@yahoo.co.uk

2. Dr Frank Baiden, MD, MSc, PhD, Centre for Health Research and Implementation Support, Ghana. Email: baidenf@yahoo.co.uk

3. Ms Barbara Araba Yankey, MPH, MPA, MSc, PhDabd. Georgia State University, Atlanta, USA. Email: hellobabs8@yahoo.com

4. Professor Kofi Awusabo-Asare, PhD, Dept of Population and Health, Faculty of Social Science, University of Cape Coast, Cape Coast Ghana. Email: awusabo_asare@yahoo.co.uk